Author’s response to reviews

Title: The association between deep learning approach and Case Based Learning

Authors:
Meenakshi Jhala (mjhala28@gmail.com)
Jai Mathur (jai@mathur.ca)

Version: 2 Date: 26 Feb 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your response and feedback.

We agree with your helpful comments and have made the specific changes that were outlined in your review of our letter. Below is more detail on what has changed:

1) The word "knowledge" has been replaced with "learning" in the first sentence of the first paragraph, and in the abstract

2) The word "methods" has been replaced with "approaches" in the third sentence of the second paragraph

3) The words "completely valid" have been changed to "understandable" in the second sentence of the fourth paragraph

4) An addition of the word "may" in the fifth sentence of the fourth paragraph

5) The words "the majority of students" have been replaced with "many students may" in the fifth sentence of the fourth paragraph

6) The addition of the word "can" in the fifth sentence of the sixth paragraph

7) The addition of the words "we believe that" in the third sentence of the seventh paragraph

We believe these changes have allowed our opinions to be outlined from facts, alongside subtle changes of wording, which has made our letter more effective.
We are very grateful for your feedback and we hope it meets the requirements specified. Please do let us know if there is anything else you require.

Yours Sincerely,

Meenakshi Jhala
1st Author